



Document History and Status

Revision	Date	Purpose/Status	File Ref	Author	Check	Review
D1	December 2015	Comment	TAMjw12066- 76-241115- D1.doc	T Marsland	S Ash	S Ash

This document has been prepared in accordance with the scope of Campbell Reith Hill LLP's (CampbellReith) appointment with its client and is subject to the terms of the appointment. It is addressed to and for the sole use and reliance of CampbellReith's client. CampbellReith accepts no liability for any use of this document other than by its client and only for the purposes, stated in the document, for which it was prepared and provided. No person other than the client may copy (in whole or in part) use or rely on the contents of this document, without the prior written permission of Campbell Reith Hill LLP. Any advice, opinions, or recommendations within this document should be read and relied upon only in the context of the document as a whole. The contents of this document are not to be construed as providing legal, business or tax advice or opinion.

© Campbell Reith Hill LLP 2015

Document Details

03/12/2015 10:50
TAMjw12066-76-241115-D1.doc
T Marsland, MEng CEng MIStructE
E M Brown, BSc MSc CGeol FGS
12066-76
26 Lyndhurst Road, London, NW3 5PB
2015/2548/P

Structural ◆ Civil ◆ Environmental ◆ Geotechnical ◆ Transportation



Contents

1.0	Non-technical summary	1
2.0	Introduction	3
3.0	Basement Impact Assessment Audit Check List	5
4.0	Discussion	9
5.0	Conclusions	10

Date: December 2015

Status: D1

Appendix

Appendix 1: Residents' Consultation Comments

Appendix 2: Audit Query Tracker Appendix 3: Supplementary Supporting Documents



1.0 NON-TECHNICAL SUMMARY

- 1.1. CampbellReith was instructed by London Borough of Camden, (LBC) to carry out an audit on the Basement Impact Assessment submitted as part of the Planning Submission documentation for 26 Lyndhurst Road, London, NW3 5PB (planning reference 2015/2548/P). The basement is considered to fall within Category B as defined by the Terms of Reference.
- 1.2. The Audit reviewed the Basement Impact Assessment for potential impact on land stability and local ground and surface water conditions arising from basement development in accordance with LBC's policies and technical procedures.
- 1.3. CampbellReith was able to access LBC's Planning Portal and gain access to the latest revision of submitted documentation and reviewed it against an agreed audit check list.
- 1.4. There is an overarching BIA report by Site Analytical Services Ltd which includes a construction method statement. There is in addition supplementary supporting information submitted in the form of the following separate reports:-
 - Existing structure assessment by Hockley & Dawson.
 - 'Factual Report on a Ground Investigation' by Site Analytical Services Ltd.
 - 'Phase 1 Preliminary Risk Assessment' by Site Analytical Services Ltd.

The authors of these documents have suitable credentials.

- 1.5. The Basement Impact Assessments at 26 Lyndhurst Road relate to the proposed development of a partially buried swimming pool in the rear garden and a new lightwell at the front of the property. Excavation depths are approximately 2.5m and 3.0m below local garden levels respectively.
- 1.6. The proposed basements will be constructed using reinforced concrete 'L' shaped underpins dowelled together and will be founded within the Made ground or Claygate member. There is a construction methodology in Appendix C of the main BIA report in word form, but no accompanying technical calculations or drawings. We would expect technical structural drawings showing the design of the retaining walls and the reinforcement within the underpins to be included with the submitted package.
- 1.7. Both basements are likely to be just above the water table level and groundwater is not expected to be encountered. If ground water is encountered it is not likely to be significant and the proposed excavation works are not considered to have an impact on the local groundwater regime.

Status: D1

Date: December 2015



- 1.8. Analysis has been undertaken of estimated horizontal and vertical ground movements based on the above construction technique and are contained within Appendix D of the main report. The analysis shows that the effect on the neighbouring properties is predicted to be very low (c2mm). This corresponds to categories 1 or 2 after Burland 1995. No proposals are provided for a movement monitoring strategy during excavation and construction, however this is not considered necessary.
- 1.9. The proposed scheme will result in a net increase in permeable surface area on site (additional 17 m² permeable areas). The area is not known to experience flooding. The BIA concludes there will be no adverse impact on surface flows from the proposed basement.



2.0 INTRODUCTION

- 2.1. CampbellReith was instructed by London Borough of Camden (LBC) on 13/10/2015 to carry out a Category B Audit on the Basement Impact Assessment (BIA) submitted as part of the Planning Submission documentation for 26 Lyndhurst Road, London, NW3 5PB (2015/2548/P).
- 2.2. The Audit was carried out in accordance with the Terms of Reference set by LBC. It reviewed the Basement Impact Assessment for potential impact on land stability and local ground and surface water conditions arising from basement development.
- 2.3. A BIA is required for all planning applications with basements in Camden in general accordance with policies and technical procedures contained within
 - Guidance for Subterranean Development (GSD). Issue 01. November 2010. Ove Arup & Partners.
 - Camden Planning Guidance (CPG) 4: Basements and Lightwells.
 - Camden Development Policy (DP) 27: Basements and Lightwells.
 - Camden Development Policy (DP) 23: Water
- 2.4. The BIA should demonstrate that schemes:
 - maintain the structural stability of the building and neighbouring properties;
 - avoid adversely affecting drainage and run off or causing other damage to the water environment; and,
 - c) avoid cumulative impacts upon structural stability or the water environment in the local area.

and evaluate the impacts of the proposed basement considering the issues of hydrology, hydrogeology and land stability via the process described by the GSD and to make recommendations for the detailed design.

- 2.5. LBC's Audit Instruction described the planning proposal as "Conversion of existing 4x flats into 1x6bed maisonette and 2x1bed flat. Alterations at rear to include single storey extension and part first floor extension; enlargement of terraces, creation of inset roof terrace and excavation for swimming pool. Alterations at front to include excavation of lightwell."
- 2.6. CampbellReith accessed LBC's Planning Portal on 19th November 2015 and gained access to the following relevant documents for audit purposes:

Status: D1

Date: December 2015



- 'Basement Impact Assessment' report by Site Analytical Services Ltd.
- Existing structure assessment by Hockley & Dawson.
- 'Factual report on a ground investigation' by Site Analytical Services Ltd.
- 'Phase 1 Preliminary Risk Assessment' by Site Analytical Services Ltd.
- Planning Application Drawings consisting of:-
- Existing and Proposed plans and sections by Carden & Godfrey Architects.



3.0 BASEMENT IMPACT ASSESSMENT AUDIT CHECK LIST

Item	Yes/No/NA	Comment	
Are Individual report (from Section 1.4) Author(s) credentials satisfactory?	Yes	Chartered Geologists, Chartered Engineers and Chartered Institute of Water and Environmental Management members.	
Is data required by Cl.233 of the GSD presented?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Does the description of the proposed development include all aspects of temporary and permanent works which might impact upon geology, hydrogeology and hydrology?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Are suitable plan/maps included?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Do the plans/maps show the whole of the relevant area of study and do they show it in sufficient detail?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Land Stability Screening: Have appropriate data sources been consulted? Is justification provided for 'No' answers?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Hydrogeology Screening: Have appropriate data sources been consulted? Is justification provided for 'No' answers?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Hydrology Screening: Have appropriate data sources been consulted? Is justification provided for 'No' answers?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	



Item	Yes/No/NA	Comment	
Is a conceptual model presented?		See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Land Stability Scoping Provided? Is scoping consistent with screening outcome?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Hydrogeology Scoping Provided? Is scoping consistent with screening outcome?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Hydrology Scoping Provided? Is scoping consistent with screening outcome?	No	Screening is sufficient.	
Is factual ground investigation data provided?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Is monitoring data presented?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Is the ground investigation informed by a desk study?	No	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Has a site walkover been undertaken?	Unknown	Not mentioned in any of the reports. Level of detail in reports suggests that a site walkover is likely to have occurred.	
Is the presence/absence of adjacent or nearby basements confirmed?	Yes	Ground Movement Assessment contained within Appendix D states that neighbouring properties Nos 25 and 27 are understood to have lower ground floors, which are taken to lie at the same level as existing lower ground floor of No. 26.	



Item	Yes/No/NA	Comment	
Is a geotechnical interpretation presented?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Does the geotechnical interpretation include information on retaining wall design?	No	Not specifically, although this document informs the Construction Method statement which does detail the retaining wall design.	
Are reports on other investigations required by screening and scoping presented?	No	Not required.	
Are baseline conditions described, based on the GSD?	Yes	To extent commensurate with scale of basement proposals.	
Do the base line conditions consider adjacent or nearby basements?	Yes	Neighbouring properties lower ground floors are discussed.	
Is an Impact Assessment provided?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Are estimates of ground movement and structural impact presented?	Yes	See Appendix D Ground Movement Assessment.	
Is the Impact Assessment appropriate to the matters identified by screen and scoping?	Yes	See accompanying overarching BIA report by 'Site Analytical Services Ltd' and accompanying documents and drawings.	
Has the need for mitigation been considered and are appropriate mitigation methods incorporated in the scheme?	No	Not considered necessary.	
Has the need for monitoring during construction been considered?	No	Not proposed however the short plus long term movements anticipated are very low (<5mm) thus movement monitoring is not thought necessary.	



Item	Yes/No/NA	Comment
Have the residual (after mitigation) impacts been clearly identified?	No	Not applicable.
Has the scheme demonstrated that the structural stability of the building and neighbouring properties and infrastructure will be maintained?	Yes	Construction Method Statement and Ground Movement Assessment are thorough well considered reports.
Has the scheme avoided adversely affecting drainage and run-off or causing other damage to the water environment?	Yes	Surface water run-off will not be increased. In fact the proposals show an increase in permeable areas of 17.6 m2.
Has the scheme avoided cumulative impacts upon structural stability or the water environment in the local area?	Yes	Yes in both areas.
Does report state that damage to surrounding buildings will be no worse than Burland Category 2?	Yes	Based on Ground Movement Assessment contained within Appendix D
Are non-technical summaries provided?	Yes	



4.0 DISCUSSION

- 4.1. The Basement Impact Assessment is an overarching report concerning all aspects of the project.

 The report authors appear to have suitable qualifications.
- 4.2. The BIA indicates that the front and rear proposed basements (approximately 3.0 2.5m bgl respectively, front basement as a lightwell and rear basement as a partially buried swimming pool) will be constructed using reinforced concrete 'L' shaped underpins dowelled together, and will be founded within the made ground or Claygate member. The exploratory holes extended to a maximum depth of 15m below ground level, with other boreholes extending to 8m below ground level.
- 4.3. Detailed structural design for the retaining wall are expected to confirm assumptions in the ground movement assessment.
- 4.4. The Ground Movement Assessment in Appendix D discusses neighbouring properties lower ground floors.
- 4.5. The conclusions reached in the Subterranean groundwater assessment is that both proposed basements are founded above the reported ground water level, achieve via testing in one of the boreholes. Whilst there is no guarantee that water table levels will not rise above the tested level, the proposed basements are not thought to have an impact on the existing groundwater regime.
- 4.6. An assessment of vertical and horizontal ground movements has been produced which estimates that the effect on neighbouring properties will be negligible.
- 4.7. No ground movement monitoring has been proposed which as the short plus long term movements to neighbouring properties are predicted to be very low is not thought necessary.
- 4.8. The Surface water assessment notes that there is a proposed increase in permeable areas.
- 4.9. It is accepted that there are no slope stability concerns regarding the proposed development and it is not in an area prone to flooding.



5.0 CONCLUSIONS

- 5.1. The Basement Impact Assessment has been carried out by an established firm of consultants using individuals who possess suitable qualifications.
- 5.2. The proposed front and rear basements (c3m below respective existing ground levels) are to be founded in the made ground or Claygate member and are predicted to be above the water table level. The deepest exploratory hole extended significantly below the proposed basement levels.
- 5.3. It is not likely that the ground water table will be encountered during basement foundation excavation, although the Construction Method Statement advices the contractor to have a contingency plan in case should the need arise.
- 5.4. The BIA states proposed basement will be constructed using reinforced concrete 'L' shaped underpins dowelled together, and will be founded within the made ground or Claygate member.
- 5.5. We would recommend that specified structural calculations and proposals for the retaining was are included.
- 5.6. Horizontal and vertical ground movement analysis predicts negligible impact (c2mm) on neighbouring properties.
- 5.7. No proposals are provided for a movement monitoring strategy during excavation and construction but for the magnitude of movements anticipated it is not thought necessary.
- 5.8. It is accepted that there are no adverse impacts on slope stability, surface water flows or flooding.



Appendix 1: Resident's Consultation Comments

None



Appendix 2: Audit Query Tracker



Audit Query Tracker

Query No	Subject	Query	Status	Date closed out
1	Stability	Lack of job specific structural design calculations and drawings for the reinforced concrete underpins.	Open	



Appendix 3: Supplementary	Supporting	Documents
---------------------------	-------------------	------------------

None